Author's response to reviews

Title: Evaluation of the Rapid Scale-up of Collaborative TB/HIV Activities in TB Facilities in Rwanda, 2005-2009

Authors:

Eric S Pevzner (ecp9@cdc.gov)
Vandebriel Greet (gv2124@columbia.edu)
Lowrance David (LowranceD@rw.cdc.gov)
Gasana Michel (michelgasana@yahoo.fr)
Finlay Alyssa (avf0@cdc.gov)

Version: 5 Date: 1 June 2011

Author's response to reviews:

Uploading a revised version of what was requested by the reviewer. The sentence proposed by the reviewer needed one additional clarifying word to be accurate. What was missing was not nationally representative data but rather nationally representative "TB/HIV" data. So, in this latest version we inserted the word TB/HIV.